Author's response to reviews

Title: The study of anthropometric estimates in the visceral fat of healthy individuals

Authors:

Chun-Hao Chen (haogoochen@gmail.com)
Yu-Yawn Chen (yu11.tw@yahoo.com.tw)
Chih-Lin Chuang (bat911888@yahoo.com.tw)
Li-Ming Chiang (chiang1999@gmail.com)
Shu-Min Chiao (smchiao@thu.edu.tw)
Kuen-Chang Hsieh (abaqus0927@yahoo.com.tw)

Version: 2
Date: 26 December 2013

Author's response to reviews: see over
Dear Editor,

Uploaded please find a manuscript entitled “The study of anthropometric estimates in the visceral fat of healthy individuals” for resubmission to *Nutrition Journal*. We have carefully revised the format according to the required form by journal. The research is original and relevant to body composition and epidemiology. We have no conflicts of interest in the present study with agreement from all authors. Please feel free to contact me for any problems. All the experiments were performed under the permission of Institutional Review Board (IRB) of Advisory Committee at Jen-Ai Hospital in Taiwan. We would like to declare that this original study has not been submitted to and not being considered to be published by any other journals. Due to the same contribution from both the Yu-Yawn Chen Ph.D. and Chun-Hao Chen, we would like have two first authors. For the sakes of time pressed to promotion of one of our author, we will very appreciate for the reviewing process as soon as possible.

Thank you very much.

Sincerely,

Kuen Chang Hsieh Ph.D.

*Corresponding author: Kuen-Chang Hsieh
Office of Physical Education and Sport,
National Chung Hsing University, 250 Kuo Kuang Rd.,
Taichung 402, Taiwan R.O.C.,
Tel: 886-4-24063766*227, Fax: 886-4-24065612
E-mail:abaqus0927@yahoo.com.tw